Correspondence

E-mail: aprilarmstrong@post.harvard.edu

Accepted for publication

April W. Armstrong.

23 September 2013

None

sen.

manuscript

Funding sources

Conflicts of interest

A.W.A. has received research grants and/or con-

sultant honoraria from Abbott, Amgen and Jans-

L.H. and B.T.S. are joint first authors of this

*Plain language summary available online

DOI 10.1111/bjd.12654

Antidrug antibodies in psoriasis: a systematic review*

L. Hsu, B.T. Snodgrass and A.W. Armstrong

Department of Dermatology, University of California, Davis, 3301 C Street, Suite 1400, Sacramento, CA 95816, U.S.A.

Summary

Antidrug antibodies (ADAs) against biological agents may be clinically significant and potentially alter a biological drug's treatment efficacy. This systematic review aims to (i) determine the prevalence of ADAs against infliximab, etanercept, adalimumab and ustekinumab in patients with psoriasis; (ii) ascertain whether ADAs are associated with changes in drug efficacy; and (iii) explore the use of concomitant methotrexate to prevent ADA formation. Through a systematic search using Medline and Embase from 29 January 1950 to 29 March 2013, we identified 25 studies that met the inclusion criteria. Of 7969 patients with psoriasis, 950 tested positive for ADAs. Antibodies against infliximab, etanercept, adalimumab and ustekinumab were reported in 5.4-43.6%, 0-18.3%, 6-45% and 3.8-6% of patients, respectively. Anti-infliximab antibodies were associated with lower serum infliximab concentrations in three studies, and decreased treatment response in five studies. ADAs against etanercept were non-neutralizing and not associated with any apparent effects on clinical response. Antiadalimumab antibodies were associated with lower serum adalimumab concentrations in three of five studies, and reduced clinical efficacy in four studies. Two of six studies reported that antiustekinumab antibodies were associated with lower Psoriasis Area and Severity Index responses, and three ustekinumab studies noted that most of these antibodies were neutralizing. Although the use of concomitant methotrexate with biological agents to prevent ADA formation in other immunemediated diseases is promising, their use in psoriasis is sparse. ADA development remains a challenge with biological therapies and therefore should be considered in patients with psoriasis who experience diminished treatment response.

What's already known about this topic?

- Antidrug antibodies have been shown to form in patients with psoriasis receiving infliximab, etanercept, adalimumab or ustekinumab.
- Certain antidrug antibodies have been noted to influence treatment efficacy.

What does this study add?

• This is among the first reviews to examine the available scientific evidence on the prevalence of antidrug antibodies against biological agents, their impact on treatment efficacy, and the utility of concomitant methotrexate to prevent antidrug antibody formation in patients with psoriasis.

Psoriasis is an immune-mediated disease characterized by well-demarcated patches and plaques on the skin; it is associated with psoriatic arthritis and other comorbidities.^{1,2} Although the pathogenesis of psoriasis is poorly understood, upregulation of T-helper (Th)1 and Th17 pathways resulting in increased levels of tumour necrosis factor (TNF)- α and

interleukin (IL)-23 has been implicated.^{3–5} Through the growing understanding of the pathogenesis of psoriasis, treatments have evolved from topical agents to biological therapies including fusion proteins and monoclonal antibodies.

Monoclonal antibodies are categorized as murine, chimeric, humanized or fully human. Murine antibodies induce antidrug

Antidrug antibodies (Al

antibodies (ADAs) against the murine variable and constant domains, limiting their therapeutic effectiveness due to significant immunogenicity.⁶ Chimeric antibodies are composed of human constant domains and murine variable domains, suggesting less immunogenicity compared with murine antibodies.^{7,8} Humanized antibodies contain murine sequences only in the antigen-binding sites. Fully human antibodies are synthesized with fully human sequences and are therefore theoretically less immunogenic.⁹

Infliximab, etanercept, adalimumab and ustekinumab are approved by the US Food and Drug Administration for treatment of moderate-to-severe psoriasis.^{10–13} Infliximab, a chimeric IgG1 κ antibody, binds to soluble and membranebound TNF- α .¹⁴ Etanercept, a fully human fusion protein composed of two p75 TNF receptors fused to a single IgG1 Fc subunit, binds to both TNF- α and TNF- β , but with less affinity for TNF- α than infliximab. Adalimumab is a fully human IgG1 κ antibody that targets both soluble and membrane-bound TNF- α .¹⁵ Ustekinumab, another fully human IgG1 κ antibody, binds the common p40 subunit of IL-12 and IL-23.¹⁶

The clinical significance of ADAs needs to be thoroughly explored, as studies suggest that they may be associated with decreased drug efficacy.^{17–19} We conducted a systematic review to (i) determine the incidence of ADA formation against infliximab, etanercept, adalimumab and ustekinumab in psoriasis; (ii) ascertain whether ADAs are associated with changes in drug efficacy in psoriasis; and (iii) determine whether concomitant methotrexate prevents ADA formation in the psoriasis population.

Materials and methods

Data sources and study selection

We performed a systematic search using the Medline and Embase databases from 29 January 1950 to 29 March 2013 with the following search query: (infliximab OR etanercept OR adalimumab OR ustekinumab) AND psoriasis AND clinical trial[ptyp] AND English[lang]. The initial search yielded 484 unique references, which were combined with six handsearched references identified from the references of relevant articles, resulting in 490 references (Fig. 1).^{20,21} For any studies involving the same patient cohort, we selected the latest primary article with the longest follow-up period.

We applied the inclusion criteria to the 490 identified references. The inclusion criteria were having at least 15 patients in the study, documenting serial skin assessments of psoriasis severity, and reporting ADAs in patients with psoriasis receiving biological agents (infliximab, etanercept, adalimumab or ustekinumab). Two authors (A.W.A. and B.T.S.) independently read the abstracts of the 490 articles and excluded 352 articles based on the selection criteria. The two authors then reviewed the full text of the remaining 138 articles and identified 25 that met the inclusion criteria. To measure study quality, we used the five-point Jadad scale for reporting randomized controlled trials (RCTs), and a six-point scale, which was previously described and adopted from a proposal by Stroup et al., for reporting observational studies.^{22–24} The Jadad scale has values of 0 or 1 assigned to the mention of randomization,



Fig 1. Selection process for study inclusion in the systematic review.

appropriateness of randomization method, mention of blinding, appropriateness of blinding method, and accounting of all patients.²³ RCTs with a score of 0–3 were categorized as lower quality and studies with a score of 4–5 were categorized as higher quality. The observational study six-point scale has values 0 or 1 assigned to study design, assessment of exposure, assessment of outcome, control for confounding, evidence of bias and assessment of psoriasis severity. Observational studies with a score of 0–3 were categorized as lower quality, and studies with a score of 4–6 were categorized as higher quality.²² The ratings measure the study quality for the primary end points such as treatment efficacy, rather than antibody formation, for most of the studies examined herein.

Data extraction

Two reviewers extracted data independently from each of the 25 articles, with discrepancies resolved by discussion (Tables 1–4). The data collected for each study included (i) study identification including year of publication; (ii) biological drug evaluated; (iii) study design; (iv) prevalence of ADAs; (v) testing modality; (vi) testing period; (vii) whether samples were drawn during trough drug levels; (viii) relationship of ADAs to psoriasis severity; (ix) methotrexate use and its relationship to ADA formation; and (x) neutralizing status of the ADAs.

ADA prevalence was determined with respect to the total number of patients tested in each study, rather than the total number of patients in the study, in order to reduce the underestimation of ADA formation. Although not all patients in the studies were tested for various reasons, this approach remained the most precise and structured interpretation of ADA prevalence for comparison with other studies. A few studies had partial information regarding the total number of patients undergoing ADA testing. Some studies reported the ADA status as inconclusive or indeterminate when both the antibody and biological drug were present, while other studies reported the ADA status dichotomously as positive or negative. If the study authors could not be contacted, we inferred the number of patients testing negative for ADAs based on the data of the total study population tested and the number of patients testing positive for ADAs.^{11,25-28}

Results

We identified 25 studies with 7969 patients with psoriasis who had laboratory tests to determine ADA status, and 950 patients tested positive for ADAs.

Anti-infliximab antibodies

Ten studies documented the prevalence of anti-infliximab antibodies (AIAs). Four were RCTs, four were prospective cohort studies and two were retrospective studies.^{13,26,29–36} AIAs were reported in 5.4-43.6% of patients (Table 1). Four studies used enzyme-linked immunosorbent assays (ELISAs) to detect AIAs, and six studies tested for AIAs at drug trough levels.^{26,29,31,32,34,36} Three studies reported an association between AIAs and lower serum infliximab concentrations.^{31,34,36} Five studies noted an association between AIAs and decreased clinical improvement.^{13,26,29,32,33} In the study of Reich et al., of those patients achieving a 75% improvement in Psoriasis Area and Severity Index (PASI 75 response) at week 10, 81% (106/131) of AIA-negative patients and 39% (20/51) of AIA-positive patients maintained this response through week 50.¹³ Krathen et al. noted that 92% (11/12) of AIA-positive patients did not maintain a 'clear' or 'almost clear' assessment on the Physician's Global Assessment scale during 1 year of infliximab therapy.³³ Adişen et al. reported that 33% of patients (five of 15) tested AIA-positive from the fifth to the thirteenth infliximab infusion.²⁹ The mean PASI scores in AIA-negative and AIA-positive patients were 5.3 and 10, respectively (P = 0.02).²⁹ Torii et al. found that 20% of patients (10/50) tested AIA-positive.³⁶ At 8 weeks postinfliximab infusion, AIA-negative and AIA-inconclusive patients had serum infliximab levels $> 0.1 \ \mu g \ mL^{-1}$, whereas AIA-positive patients had undetectable levels (< $0.1 \ \mu g \ mL^{-1}$).³⁶ The PASI 75 response rates were 96% (22/23) in patients with serum infliximab concentrations of 1–10 μ g mL⁻¹, 71% (five out of seven) in patients with serum infliximab concentrations of $0.1-1 \ \mu g \ mL^{-1}$, and 60% (six out of 10) in patients with serum infliximab concentrations $< 0.1 \ \mu g \ mL^{-1}$ (P-value not specified).³⁶ Hoffmann et al. reported AIAs in 21% of patients (six out of 29).³² The authors noted worsening skin or joint symptoms in 13% of AIA-negative patients (three out of 23) and four of six AIA-positive patients.³² Takahashi et al. reported AIAs in 30% of patients (six out of 20) at 48 weeks.³⁴ The authors reported 48-week mean serum trough infliximab concentrations of $4.12 \ \mu g \ mL^{-1}$ in 14 ADA-negative patients and $1.21 \ \mu g \ mL^{-1}$ in six ADA-positive patients (P < 0.01).³⁴ No articles reported the neutralizing status of AIAs.

Three infliximab studies reported data on the relationship of methotrexate with ADA incidence.^{29,32,33} Krathen *et al.* detected AIAs in 10/66 patients (15%) without concomitant methotrexate and in two of seven patients receiving a mean methotrexate dose of 7.5 mg per week.³³ Adişen *et al.* found that an unspecified number of the five AIA-positive patients became AIA negative after 8 weeks of methotrexate 5–15 mg per week.²⁹ Hoffman *et al.* reported AIAs in five of 22 patients (23%) without concomitant methotrexate and in one of seven patients treated with an unspecified methotrexate dose.³²

Antietanercept antibodies

Six studies measured antietanercept antibodies (AEAs) in patients with psoriasis. Four were RCTs and two were prospective cohort studies.^{11,25,37–40} The prevalence of AEAs ranged from 0% to 18·3% (Table 2). Five studies detected AEA status with ELISA, and one study documented collecting serum samples while simultaneously collecting the serum drug trough samples. Leonardi *et al.* reported that the percentages of AEA-negative patients with a PASI 75 response at weeks 48

Quality rating (observational studies) ^b	1 1	4	I	v
Quality rating (Jadad scale) ^a	ν ν	1	Ŋ	I
Methotrexate use	Not used Not used	Two of seven patients on methotrexate at 5–7.5 mg per week developed AIAs	Not used	An unspecified number of AIA-positive patients tested negative for AIAs after 8 weeks of methotrexate
Relation to clinical efficacy	Not specified Of PASI 75 responders at week 10, 81% (106) of AlA-negative, 96% (22) of AlA-inconclusive and 39% (20) of AlA-positive patients maintained a PASI 75 response at week 50 (<i>P</i> -value nor reported)	92% (11/12) of AIA-positive patients had loss of efficacy in < 1 year	AIA-positive patients were less likely to maintain response through week 50 than AIA-negative patients (no. patients and P-value not specified)	Mean PASI scores in AIA-negative and AIA-positive patients were 5-3 and 10, respectively (P = 0.02)
Collected at trough	Not reported Not reported	Not reported	Yes	Yes
Testing modality	ELISA Not reported	Not reported	Not reported	HIISA
Testing period	To week 26 At weeks 0, 46, 66	At or beyond month 12, only those who lost response were tested	At weeks 0, 30, 50 and 66	To 84 weeks
Antidrug antibody prevalence	23.3% (38/163) 26.5% (70/264)	16% (12/73) ^c	43.6% (237/543)°	33% (5/15)
Study design	RCT RCT	Retrospective study	RCT	Prospective cohort study
Study	Gottlieb 2004 ³⁰ Reich 2005 ¹³	Krathen 2006 ³³	Menter 2007 ²⁶	Adişen 2010 ²⁹

Table 1 Anti-infliximab antibodies (AIAs) in psoriasis

Study	Study design	Antidrug antibody prevalence	Testing period	Testing modality	Collected at trough	Relation to clinical efficacy	Methorrexate use	Quality rating (Jadad scale) ^a	Quality rating (observational studies) ^b
Torii 2010 ³⁶	RCT	20% (10/50)	To week 78	Not reported	Yes	8 weeks postinfliximab infusion, AlA-negative and AlA-inconclusive patients had serum infliximab levels > 0.1 µg mL ⁻¹ ; AlA-positive patients had undetectable levels (< 0.1 µg mL ⁻¹). At week 62, PASI 75 response rates were 96% (22/23) in patients with serum infliximab 1–10 µg mL ⁻¹ , 71% (5/7) for 0.1–1 µg mL ⁻¹ and 60% (6/10) for < 0.1 µg mL ⁻¹	Not used	4	1
Hoffmann 2011 ^{3 2}	Retrospective case–control study	21% (6/29)	After a median of 11 infusions (range 3–21)	ELISA	Yes	Four of six AIA-positive patients had worsening of skin or joint symptoms vs. 13% (3/23) of AIA-negative patients (P-value not specified)	One of seven patients on an unspecified dose of methotrexate had AIAs	I	v
Torii 2011 ³⁵	Prospective cohort study	30% (19/64)	To week 54	Not reported	Not reported	Not specified	Four of 64 patients received methotrexate, but AIA status was not reported	I	S
Gottlieb 2012 ³¹	Prospective cohort study	5.4% (9/168)	At week 22	Not reported	Yes	Serum infliximab trough concentrations at week 22 were undetectable in an unspecified number of AIA-positive patients (<i>p</i> -value not specified)	Eight patients received methotrexate at doses < 25 mg per week. No report on the relationship of methotrexate to AIAs	1	ý

Study	Antidrug an Study design prevalence	Antidrug antibody prevalence	Testing period	Testing modality	Collected at trough	Relation to clinical efficacy	Methotrexate use	rating (Jadad scale) ^a	Quality rating (observational studies) ^b
Takahashi 2013 ³⁴ Prospective cohort study	Prospective cohort study	30% (6/20)	At week 48	ELISA	Yes	AIA-positive patients had lower serum trough infliximab concentrations than AIA-negative patients (P < 0.01)	Not used	I	N

and 72 were 52% and 49%, respectively.³⁹ Those who tested AEA-positive had comparable response rates to etanercept with those testing negative. Specifically, 52% and 49% of patients who tested negative for AEAs achieved a PASI 75 response at weeks 48 and 72, respectively; 54% and 57% of patients testing AEA-positive three or more times achieved a PASI 75 response at weeks 48 and 72, respectively.³⁹ Tyring *et al.* reported PASI 75 responses of 64% and 60% in AEA-negative and AEA-positive patients, respectively, at week 12.²⁵ Three studies found no apparent difference in safety or efficacy between AEA-negative and AEA-positive patients.^{11,37,38} All AEAs were reported as non-neutralizing. Concomitant methotrexate was not used in the six studies.

Antiadalimumab antibodies

Five studies reported antiadalimumab antibodies (AAAs) in patients with psoriasis.^{12,27,34,40,41} Two studies were RCTs and three were prospective cohort studies. These studies described incidence rates of AAAs from 6% to 45% (Table 3). Three studies utilized ELISA to detect AAAs, 27, 34, 40 and one study used radioimmunoassay (RIA).⁴¹ Three of the studies reported collecting samples immediately prior to administration of the next drug dose.^{27,34,41} Takahashi et al. and Lecluse et al. reported higher intravascular adalimumab concentrations in AAA-negative patients than in AAA-positive patients.^{34,41} Four of the five studies reported decreased drug efficacy with AAA positivity.^{12,34,40,41} Menter et al. reported AAAs in 8.8% of patients (73/825).¹² Of the patients who achieved a PASI 75 response by week 33, 27.9% (65/233) of AAA-negative patients and 42.8% of AAA-positive patients lost clinical response (PASI < 50) by week 52.¹² Asahina et al. documented AAAs in 10.6% of patients (13/123).²⁷ At week 16, 72.7% of AAA-negative patients achieved at least a PASI 75 response, compared with 23.1% of AAA-positive patients (P < 0.001).²⁷ Lecluse et al. reported week-24 median adalimumab trough concentrations of 9.6, 1.3 and 0.0 mg L^{-1} in AAA-negative, AAA-positive low-titre and AAA-positive high-titre patients, respectively (P < 0.001).⁴¹ At week 24, the PASI 75 response rate decreased as AAA titres increased. The authors documented that a PASI 75 response was achieved by 56% of patients (nine out of 16) without antidrug antibodies, one of six patients with a low titre of AAAs, and none of the seven patients with a high titre of AAAs (P < 0.001).⁴¹ Takahashi et al. measured mean trough adalimumab concentrations of 7.27 $\mu g \ m L^{-1}$ in 28 AAA-negative patients vs. 2.77 μ g mL⁻¹ in five AAA-positive patients at week 48 (P < 0.01).³⁴ No articles reported the neutralizing status of the AAAs. Lecluse et al. found AAAs in 50% of patients (13/26) not on methotrexate and in none of three patients receiving concomitant methotrexate at a mean dose of 12 mg per week.⁴¹

Antiustekinumab antibodies

Six studies reporting antiustekinumab antibodies (AUAs) were examined. Five were RCTs and one was a prospective cohort

Table 1 (continued)

Study	Study design	Antidrug antibody prevalence	Testing period	Testing modality	Collected at trough	Relation to clinical efficacy	Methotrexate use	Quality rating (Jadad scale) ^a	Quality rating (observational studies) ^b
Leonardi 2003 ¹¹	RCT	Eight patients positive; negative patients not	At week 24	ELISA	Not reported	No apparent difference (P-value not specified)	Not used	4	1
Papp 2005 ³⁷	RCT	specified 2.7% (15/549)	At weeks 12, 24 or	ELISA	Not reported	No apparent difference	Not used	5	I
Gordon 2006 ³⁸	RCT	4.7% (14/297)	Up to at least 60 weeks	ELISA	Not reported	(r-vaue not spectred) No apparent difference (P-value not enerified)	Not used	5	I
Tyring 2007 ²⁵	RCT	18·3% (111/606) ^c	By 96 weeks	ELISA	Not reported	No apparent alteration	Not used	5	I
Leonardi 2010 ³⁹	Ч	15.2% (130/857)	By 72 weeks	Not reported	Not reported	PASI 75 response	Not used	I	4
	cohort study					at weeks 48 (5.2%) and 72 (49%) in AEA-negative patients vs. 5.2% and 48% of patients testing AEA- positive 1–2 times and 54% and 57% of patients testing positive ≥ 3 times, respectively (P-value			
Mahil 2013 ⁴⁰	Prospective cohort study	0% (0/25)	Not specified	ELISA	Not reported	not specified) No AEAs reported	Not used	I	5
RCT, randomized quality. ^b Studies v	controlled trial; with a score of 0-	ELISA, enzyme-linked imr -3 are categorized as lowe	nunosorbent assay; PA r quality and 4–6, hig	SI, Psoriasis Area ther quality. ^c De	a and Severity In nominator calcul	RCT, randomized controlled trial; ELISA, enzyme-linked immunosorbent assay; PASI, Psoriasis Area and Severity Index. "Studies with a score of 0–3 are categorized as lower quality and 4–5, higher quality. ^b Studies with a score of 0–3 are categorized as lower quality and 4–6, higher quality. [°] Denominator calculated based on the percentage and numerator given.		zed as lower qua given.	lity and 4–5, higher

Table 2 Antietanercept antibodies (AEAs) in psoriasis

Study	Study design	Antidrug antibody prevalence	Testing period	Testing modality	Collected at trough	Relation to clinical efficacy	Methotrexate use	Quality rating (Jadad scale) ^a	Quality rating (observational studies) ^b
Menter 2008 ¹²	RCT	8·8% (73/825)	Once or more during 52 weeks and at day 70	Not reported	Not reported	Three of seven AAA-positive patients and 65 (27.9%) of 233 AAA-negative patients lost adequate response	Not used	2	1
Lechuse 2010 ⁴¹	Prospective cohort study	45% (13/29)	At weeks 12 and 24	RIA	Yes	At week 24, median adalimumab trough concentrations in AAA-negative, AAA-positive low-titre and AAA-positive high-titre patients were 9-6, 1.3 and 0-0 mg L ⁻¹ , respectively; ms. patients with PASI 50 response were 81% (13/16), 50% (3/6) and 0% (0/7) and PASI 75 response 56% (9/16), 17% (1/6) and 0% (0/7), respectively ($P < 0.001$)	None of 3 patients on methotrexate (mean dose 12 mg per week) developed AAAs	1	کر ا
Asahina 2010 ²⁷	RCT	10-6% (13/123) ^c	At weeks 0, 16 and 24; follow-up visits	ELISA	Yes	PASI 50, 75 and 90 responses in AAA-negative vs. AAA-positive patients were $87\cdot3\%$, $72\cdot7\%$ and $51\cdot8\%$ vs. $38\cdot5\%$, $23\cdot1\%$ and 0% , respectively ($P < 0.001$)	Not used	Ŋ	1
Takahashi 2013 ³⁴	Prospective cohort study	16% (5/32)	At week 48	ELISA	Yes	Mean trough adalimmab levels in AAA-negative and AAA-positive patients were 7.27 and $2.77 \ \mu g \ m L^{-1}$, respectively ($P < 0.01$)	Not used	T	Ś
Mahil 2013 ⁴⁰	Prospective cohort study	6% (2/31)	Not specified	ELISA	Not reported	None of the 23 patients with PASI 75 response by 6 months had AAAs. Two of eight patients without PASI 50 response by 6 months were AAA-positive	Not used	T	S

Table 3 Antiadalimumab antibodies (AAAs) in psoriasis

Study	Study design	antibody prevalence	Testing period	Testing modality	Collected at trough	Relation to clinical efficacy	Neutralizing status	Quality rating (Jadad scale) ^a	Quanty raung (observational studies) ^b
Kauffinan 2004 ¹⁶	Prospective cohort study	6% (1/18)	At week 16	ELISA	Not reported	Not specified	AUAs were neutralizing	1	5
Krueger 2007 ⁴²	RCT	4.1% (12/293)	To 52 weeks	ELISA	Excess drug removed prior to testing	Not specified	Not specified	Ŋ	1
Papp 2008 ⁴³	RCT	5.4% (65/1202) To week	To week 52	ELISA	Not reported	AUAs were present in 2.0% (12/589) and 12.7% (20/158) of PASI 75 and PASI 50 responders, respectively (P-value	Most AUAs were neutralizing	رم ا	1.
Griffiths 2010 ⁴⁴ Tsai 2011 ²⁸	RCT RCT	3.8% (32/835) 4.4% (5/113) ^c	By week 64 At week 36	Not reported Not reported	Not reported Not reported	Not specified At week 28, PASI 75 responses in AUA-negative and AUA-positive patients were 74-5% (79/106) and 60% (3/5),	Not specified Not specified	4 2	1 1
Kimball 2013 ¹⁰	RCT	5.2% (39/746)	To week 264	ELISA	Excess drug removed prior to testing	respectively (P-value not reported) No affect on clinical response. At the week-244 assessment, 4.3% (6/140) of patients lost response in the absence of AUAs and without a decrease in serum ustekinumab (P-value not reported)	64% of AUAs were neutralizing	ى ب	1

Table 4 Antiustekinumab (AUAs) antibodies in psoriasis

study.^{10,16,28,42–44} The prevalence of AUAs ranged from 3.8% to 6% (Table 4). Four studies described AUA detection by ELISA, but no studies designated whether samples were collected at trough drug levels. However, two studies reported washing sera of the drug prior to testing samples for AUAs.^{10,42} Two studies demonstrated an association between AUAs and lower PASI response.^{28,43} Papp et al. reported AUAs in 5.4% (65/1202) of patients and documented that most of the AUAs (proportion not specified) were neutralizing.⁴³ The authors also noted that 12.7% (20/158) of PASI 50 responders were AUA-positive, compared with 2.0% (12/589) of PASI 75 responders, suggesting a trend for decreased treatment response with AUA formation (P-value not reported).⁴³ Tsai et al. reported AUAs in 4.4% of patients (five out of 113).²⁸ At week 28, 79 (74.5%) of 106 AUA-negative patients achieved a PASI 75 response vs. three of five AUA-positive patients.²⁸ Concomitant methotrexate was not used in these studies.

Discussion

This is among the first systematic reviews on the prevalence of ADAs, their relationship to clinical response, and the effect of concomitant methotrexate therapy in psoriasis. Based on the studies reviewed herein, we found great variability in the rate of ADA detection in psoriasis, and this was also observed in other systemic inflammatory diseases such as rheumatoid arthritis (RA), Crohn disease (CD) and ankylosing spondylitis (AS) (Table 5). It is believed that treatment- and patientrelated factors including genetics, pharmacokinetics, disease type and drug dosing intervals may contribute to ADA formation. ADAs have been correlated with patients who had higher baseline disease activity, longer disease duration, more severe disease and increased C-reactive protein.¹⁷

Additional variables that may have contributed to the wide range of detection include the use of different ADA detection assays and timing of serum sample collection in relation to drug administration. ELISA and RIA are two commonly used assays for serum ADA detection. Although specific, ELISA is prone to false negatives because it does not detect IgG4. It also requires both Fab arms of the antibody to be bound to the assay for detection, leading to drug interference if ADAs are bound in immune complexes with the biological agent.⁴⁵ The RIA detects more clinically relevant antibodies but has been associated with false positive results.^{17,46} A majority of the studies utilized ELISA to detect ADAs, and these assays may be

Table 5 Antidrug antibody formation in other systemic inflammatory diseases

	Rheumatoid arthritis	Crohn disease	Ankylosing spondylitis
Infliximab ^{47,59,62}	12-43%	6-73%	3.0-29.0%
Etanercept ^{18,47,62,63}	3-5.6%	-	0%
Adalimumab ^{47,62,64}	1-17%	2.6-20.0%	31%
Ustekinumab ⁵⁶	_	0.7%	_

limited by the presence of the drug in the serum samples.⁴⁶ ADA development and detection are likely influenced by multiple factors. Further studies are needed to understand better their pathophysiology along with improvement in methods for optimal ADA detection.

Although ADAs were detected at variable rates among different diseases, their effects on treatment response have been shown to be relatively analogous among these diseases.^{18,47,48} ADAs have been described as neutralizing or non-neutralizing. Neutralizing antibodies are thought to interfere with the biological agent's binding activity, leading to subsequent diminished clinical response; non-neutralizing antibodies do not interfere with drug-target binding and have no effect on treatment response.9,49 AIA positivity has been associated with lower serum infliximab concentrations and diminished clinical response in psoriasis, as well as in RA, CD and AS.^{26,31,35,36,48,50} No studies of patients with psoriasis, or patients with RA or AS receiving etanercept, have shown any clear association between AEAs and reduced treatment efficacy.11,25,37-39,47,51,52 Among patients with psoriasis treated with adalimumab, AAA development was associated with lower serum adalimumab levels and lower PASI 75 responses.^{12,27,34,40,41} Studies of patients with RA have also described lower serum adalimumab levels and significantly fewer treatment responders associated with AAAs.^{47,53,54} For example, Bartelds et al. documented median adalimumab concentrations of 1.2 and 11.0 mg L⁻¹ in AAA-positive and AAAnegative patients, respectively (P < 0.001).⁵³ The authors also reported that 4% of AAA-positive patients (three out of 76) vs. 34% (67/196) of AAA-negative patients sustained disease remission at week 28 (P = 0.001).⁵³ The effects of AUAs on serum ustekinumab levels and treatment response are unclear. Researchers have reported that some AUAs were neutralizing an association with and suggested diminished response.^{10,16,28,42,43} Their impact on the treatment of other immune-mediated diseases was not reported.55,56 The results herein suggest that ADAs against certain biological agents may modify serum drug levels and influence treatment efficacy.

Different theories have been proposed to explain the clinical effects of ADAs. Structural components of biological agents have been proposed as causes of immunogenicity. Infliximab contains murine components at the drug target's binding sites, likely inducing an antigenic reaction specific to them.⁹ In contrast, the binding sites of etanercept are naturally occurring type II (p75) receptors, suggesting immunogenicity against other portions that do not compromise etanercept's therapeutic activity.9 One study reported that the antiadalimumab response in vitro was specifically confined to the TNF- α binding region of adalimumab, leading to functional neutralization of the drug.49 Another study speculated that ADAs may form immune complexes with the drug, subsequently accelerating drug and ADA clearance given the shorter half-lives of immune complexes compared with free-standing antibodies.⁵⁷ The mechanisms behind the neutralizing consequences of ADAs are not clearly understood, but their effects on treatment response are clinically significant.

To prevent potentially decreased response due to ADAs, studies have investigated the use of concomitant methotrexate to optimize treatment response. Among patients with psoriasis, concomitant methotrexate therapy with infliximab or adalimumab has been limited, although the findings suggest favourable results. Methotrexate incorporation into infliximab or adalimumab treatment of RA or CD has been more extensively studied. In a cohort of 101 patients with RA, Maini et al. showed that AIAs developed respectively in 15%, 7% or 0% of patients receiving infliximab at 1, 3 or 10 mg kg⁻¹ doses, as well as methotrexate 7.5 mg per week.⁵⁸ In CD, one study noted AIA incidences of 44% (22/50) in patients receiving concomitant methotrexate vs. 73% in patients receiving infliximab monotherapy.⁵⁹ With respect to adalimumab therapy, Weinblatt et al. reported that < 1% of patients (two of 209) with RA on both adalimumab and methotrexate developed AAAs.⁶⁰ Additionally, one study reported that 84% (84/100) of AAA-negative patients received concomitant methotrexate vs. 52% (11/21) of AAA-positive patients, indicating a decreased prevalence of immunogenicity against adalimumab with concomitant immunosuppressive agents (P = 0.003).⁵³ Methotrexate use has not been shown to alter the safety profile of biological treatments in studies with rheumatology patients.45 Concomitant methotrexate in psoriasis is limited, but its use with biological agents in other inflammatory diseases is seen optimistically.

Although concomitant methotrexate has been associated with less ADA formation in patients receiving infliximab or adalimumab, the mechanism behind this effect has not been elucidated. Studies have proposed that methotrexate acts through a synergistic or anti-immunological effect based on the magnitude of clinical response relative to monotherapy. $^{\rm 45,59,61}$ The methotrexate concentration administered to patients varied among different studies. Methotrexate coadministration of up to 25 mg weekly with biological agents is commonly used in RA, which may influence the prevalence of ADA formation in RA studies. In CD, supplementary methotrexate use is less defined, and the drug is generally administered on a short-term basis. There are no clear guidelines for prescribing or dosing concomitant immunomodulators.45 Further studies are needed to determine the optimal concomitant methotrexate dose and duration to reduce ADA formation while minimizing methotrexate toxicity. Data on methotrexate coadministration in RA and CD provide a relevant background for investigating its use with biologics for moderate-to-severe psoriasis. Methods to prevent or counteract ADA development will become increasingly crucial as the use of biological agents for psoriasis continues to grow.

The results of this systematic review need to be interpreted in the context of the primary studies. Inclusion and exclusion criteria may have excluded certain studies, potentially resulting in selection bias that may contribute to differing views on the topics addressed herein. Also, a majority of studies detected ADAs as a secondary end point, and most studies were not powered to detect significant differences among ADA-positive patients. ADA testing in patients who withdrew prematurely from studies and the timing of serum sample collection were variably reported among the studies. Additionally, ADA testing at different intervals and after receiving varying numbers of treatment infusions may influence the ADA prevalence documented in these studies. These factors subsequently lead to possible underestimation of the true prevalence, as those who were not tested may have been ADA positive. Despite these limitations, a majority of studies reported similar conclusions on ADA formation against each biological agent and their respective impacts on treatment efficacy.

In summary, biological agents have been reported to have a favourable benefit-risk ratio in clinical trials. Their use in moderate-to-severe psoriasis will continue to grow, and ADA development will remain a challenge. This is among the first systematic reviews that have examined the incidence of ADA formation and its subsequent effect on therapeutic response to infliximab, etanercept, adalimumab and ustekinumab in patients with psoriasis. Based on the studies reviewed herein, ADAs have been linked to decreased treatment efficacy with infliximab and adalimumab but not with etanercept. The effect of AUAs on treatment response is yet to be determined. Although studies of other inflammatory diseases have reported promising results with concomitant methotrexate use to prevent ADA formation, the dearth of studies examining its use in patients with psoriasis suggests a need for further investigation to determine its utility in this population. Optimizing treatment response remains the primary objective of biological therapies in psoriasis. Therefore, ADAs should be considered an important contributing factor in patients with diminished clinical response to biological agents.

References

- 1 Kurd SK, Gelfand JM. The prevalence of previously diagnosed and undiagnosed psoriasis in US adults: results from NHANES 2003– 2004. J Am Acad Dermatol 2009; 60:218–24.
- 2 Armstrong AW, Harskamp CT, Ledo L et al. Coronary artery disease in patients with psoriasis referred for coronary angiography. *Am J Cardiol* 2012; **109**:976–80.
- 3 Lee E, Trepicchio WL, Oestreicher JL et al. Increased expression of interleukin 23 p19 and p40 in lesional skin of patients with psoriasis vulgaris. J Exp Med 2004; 199:125–30.
- 4 Ettehadi P, Greaves MW, Wallach D et al. Elevated tumour necrosis factor- α (TNF- α) biological activity in psoriatic skin lesions. Clin Exp Immunol 1994; **96**:146–51.
- 5 Mussi A, Bonifati C, Carducci M et al. Serum TNF- α levels correlate with disease severity and are reduced by effective therapy in plaque-type psoriasis. J Biol Regul Homeost Agents 1997; **11**:115–18.
- 6 Saravolatz LD, Wherry JC, Spooner C et al. Clinical safety, tolerability, and pharmacokinetics of murine monoclonal antibody to human tumor necrosis factor-α. J Infect Dis 1994; **169**:214–17.
- 7 Boulianne GL, Hozumi N, Shulman MJ. Production of functional chimaeric mouse/human antibody. Nature 1984; **312**:643–6.
- 8 LoBuglio AF, Wheeler RH, Trang J et al. Mouse/human chimeric monoclonal antibody in man: kinetics and immune response. Proc Natl Acad Sci U S A 1989; 86:4220–4.
- 9 Harding FA, Stickler MM, Razo J, DuBridge RB. The immunogenicity of humanized and fully human antibodies: residual immunogenicity resides in the CDR regions. MAbs 2010; 2:256–65.

- 10 Kimball AB, Papp KA, Wash Y et al. Long-term efficacy of ustekinumab in patients with moderate-to-severe psoriasis treated for up to 5 years in the PHOENIX 1 study. J Eur Acad Dermatol Venereol 2013; 27:1535–45.
- 11 Leonardi CL, Powers JL, Matheson RT et al. Etanercept as monotherapy in patients with psoriasis. N Engl J Med 2003; 349:2014-22.
- 12 Menter A, Tyring SK, Gordon K et al. Adalimumab therapy for moderate to severe psoriasis: a randomized, controlled phase III trial. J Am Acad Dermatol 2008; 58:106–15.
- 13 Reich K, Nestle FO, Papp K et al. Infliximab induction and maintenance therapy for moderate-to-severe psoriasis: a phase III, multicentre, double-blind trial. Lancet 2005; 366:1367-74.
- 14 Knight DM, Trinh H, Le J et al. Construction and initial characterization of a mouse-human chimeric anti-TNF antibody. Mol Immunol 1993; 30:1443–53.
- 15 Mitoma H, Horiuchi T, Tsukamoto H et al. Mechanisms for cytotoxic effects of anti-tumor necrosis factor agents on transmembrane tumor necrosis factor α -expressing cells: comparison among infliximab, etanercept, and adalimumab. *Arthritis Rheum* 2008; **58**:1248–57.
- 16 Kauffman CL, Aria N, Toichi E et al. A phase I study evaluating the safety, pharmacokinetics, and clinical response of a human IL-12 p40 antibody in subjects with plaque psoriasis. J Invest Dermatol 2004; 123:1037–44.
- 17 Bartelds GM, Krieckaert CL, Nurmohamed MT et al. Development of antidrug antibodies against adalimumab and association with disease activity and treatment failure during long-term follow-up. JAMA 2011; 305:1460–8.
- 18 Garces S, Demengeot J, Benito-Garcia E. The immunogenicity of anti-TNF therapy in immune-mediated inflammatory diseases: a systematic review of the literature with a meta-analysis. Ann Rheum Dis 2013; 72:1947–55.
- 19 Baert F, Noman M, Vermeire S et al. Influence of immunogenicity on the long-term efficacy of infliximab in Crohn's disease. N Engl J Med 2003; 348:601–8.
- 20 Brezinski EA, Armstrong AW. Off-label biologic regimens in psoriasis: a systematic review of efficacy and safety of dose escalation, reduction, and interrupted biologic therapy. PLOS ONE 2012; 7: e33486.
- 21 Sivamani RK, Goodarzi H, Garcia MS et al. Biologic therapies in the treatment of psoriasis: a comprehensive evidence-based basic science and clinical review and a practical guide to tuberculosis monitoring. Clin Rev Allergy Immunol 2013; 44:121–40.
- 22 Armstrong AW, Harskamp CT, Armstrong EJ. Psoriasis and the risk of diabetes mellitus: a systematic review and meta-analysis. JAMA Dermatol 2013; 149:84–91.
- 23 Jadad AR, Moore RA, Carroll D et al. Assessing the quality of reports of randomized clinical trials: is blinding necessary? Control Clin Trials 1996; 17:1–12.
- 24 Stroup DF, Berlin JA, Morton SC et al. Meta-analysis of observational studies in epidemiology: a proposal for reporting. Metaanalysis Of Observational Studies in Epidemiology (MOOSE) group. JAMA 2000; 283:2008–12.
- 25 Tyring S, Gordon KB, Poulin Y et al. Long-term safety and efficacy of 50 mg of etanercept twice weekly in patients with psoriasis. *Arch Dermatol* 2007; **143**:719–26.
- 26 Menter A, Feldman SR, Weinstein GD et al. A randomized comparison of continuous vs. intermittent infliximab maintenance regimens over 1 year in the treatment of moderate-to-severe plaque psoriasis. J Am Acad Dermatol 2007; 56:e1–15.
- 27 Asahina A, Nakagawa H, Etoh T et al. Adalimumab in Japanese patients with moderate to severe chronic plaque psoriasis: efficacy

and safety results from a Phase II/III randomized controlled study. J Dermatol 2010; 37:299-310.

- 28 Tsai TF, Ho JC, Song M et al. Efficacy and safety of ustekinumab for the treatment of moderate-to-severe psoriasis: a phase III, randomized, placebo-controlled trial in Taiwanese and Korean patients (PEARL). J Dermatol Sci 2011; 63:154–63.
- 29 Adişen E, Aral A, Aybay C, Gürer MA. Anti-infliximab antibody status and its relation to clinical response in psoriatic patients: a pilot study. J Dermatol 2010; 37:708-13.
- 30 Gottlieb AB, Evans R, Li S et al. Infliximab induction therapy for patients with severe plaque-type psoriasis: a randomized, doubleblind, placebo-controlled trial. J Am Acad Dermatol 2004; 51:534–42.
- 31 Gottlieb AB, Kalb RE, Blauvelt A et al. The efficacy and safety of infliximab in patients with plaque psoriasis who had an inadequate response to etanercept: results of a prospective, multicenter, openlabel study. J Am Acad Dermatol 2012; 67:642–50.
- 32 Hoffmann JH, Hartmann M, Enk AH, Hadaschik EN. Autoantibodies in psoriasis as predictors for loss of response and anti-infliximab antibody induction. Br J Dermatol 2011; 165:1355–8.
- 33 Krathen RA, Berthelot CN, Hsu S. Sustained efficacy and safety of infliximab in psoriasis: a retrospective study of 73 patients. J Drugs Dermatol 2006; 5:251–4.
- 34 Takahashi H, Tsuji H, Ishida-Yamamoto A, Iizuka H. Plasma trough levels of adalimumab and infliximab in terms of clinical efficacy during the treatment of psoriasis. J Dermatol 2013; **40**:39–42.
- 35 Torii H, Nakagawa H, Japanese Infliximab Study Investigators. Long-term study of infliximab in Japanese patients with plaque psoriasis, psoriatic arthritis, pustular psoriasis and psoriatic erythroderma. J Dermatol 2011; 38:321–34.
- 36 Torii H, Nakagawa H, Japanese Infliximab Study Investigators. Infliximab monotherapy in Japanese patients with moderate-to-severe plaque psoriasis and psoriatic arthritis. A randomized, double-blind, placebo-controlled multicenter trial. J Dermatol Sci 2010; 59:40–9.
- 37 Papp KA, Tyring S, Lahfa M et al. A global phase III randomized controlled trial of etanercept in psoriasis: safety, efficacy, and effect of dose reduction. Br J Dermatol 2005; **152**:1304–12.
- 38 Gordon KB, Gottlieb AB, Leonardi CL et al. Clinical response in psoriasis patients discontinued from and then reinitiated on etanercept therapy. J Dermatolog Treat 2006; 17:9–17.
- 39 Leonardi C, Strober B, Gottlieb AB et al. Long-term safety and efficacy of etanercept in patients with psoriasis: an open-label study. J Drugs Dermatol 2010; 9:928–37.
- 40 Mahil SK, Arkir Z, Richards G et al. Predicting treatment response in psoriasis using serum levels of adalimumab and etanercept: a single-centre, cohort study. Br J Dermatol 2013; **169**:306–13.
- 41 Lecluse LL, Driessen RJ, Spuls PI et al. Extent and clinical consequences of antibody formation against adalimumab in patients with plaque psoriasis. Arch Dermatol 2010; **146**:127–32.
- 42 Krueger GG, Langley RG, Leonardi C et al. A human interleukin-12/23 monoclonal antibody for the treatment of psoriasis. N Engl J Med 2007; 356:580–92.
- 43 Papp KA, Langley RG, Lebwohl M et al. Efficacy and safety of ustekinumab, a human interleukin-12/23 monoclonal antibody, in patients with psoriasis: 52-week results from a randomised, double-blind, placebo-controlled trial (PHOENIX 2). Lancet 2008; 371:1675–84.
- 44 Griffiths CE, Strober BE, van de Kerkhof P et al. Comparison of ustekinumab and etanercept for moderate-to-severe psoriasis. N Engl J Med 2010; 362:118–28.
- 45 Krieckaert CL, Bartelds GM, Lems WF, Wolbink GJ. The effect of immunomodulators on the immunogenicity of TNF-blocking ther-

apeutic monoclonal antibodies: a review. Arthritis Res Ther 2010; 12:217.

- 46 Hart MH, de Vrieze H, Wouters D et al. Differential effect of drug interference in immunogenicity assays. J Immunol Methods 2011; 372:196–203.
- 47 Emi Aikawa N, de Carvalho JF, Artur Almeida Silva C, Bonfá E. Immunogenicity of anti-TNF-α agents in autoimmune diseases. Clin Rev Allergy Immunol 2010; 38:82–9.
- 48 Wolbink GJ, Aarden LA, Dijkmans BA. Dealing with immunogenicity of biologicals: assessment and clinical relevance. Curr Opin Rheumatol 2009; 21:211–15.
- 49 van Schouwenburg PA, van de Stadt LA, de Jong RN et al. Adalimumab elicits a restricted anti-idiotypic antibody response in autoimmune patients resulting in functional neutralisation. Ann Rheum Dis 2013; 72:104–9.
- 50 Svenson M, Geborek P, Saxne T, Bendtzen K. Monitoring patients treated with anti-TNF- α biopharmaceuticals: assessing serum infliximab and anti-infliximab antibodies. Rheumatology (Oxford) 2007; **46**:1828–34.
- 51 Dore RK, Mathews S, Schechtman J et al. The immunogenicity, safety, and efficacy of etanercept liquid administered once weekly in patients with rheumatoid arthritis. Clin Exp Rheumatol 2007; 25:40–6.
- 52 Keystone EC, Schiff MH, Kremer JM et al. Once-weekly administration of 50 mg etanercept in patients with active rheumatoid arthritis: results of a multicenter, randomized, double-blind, placebo-controlled trial. Arthritis Rheum 2004; **50**:353–63.
- 53 Bartelds GM, Wijbrandts CA, Nurmohamed MT et al. Clinical response to adalimumab: relationship to anti-adalimumab antibodies and serum adalimumab concentrations in rheumatoid arthritis. *Ann Rheum Dis* 2007; **66**:921–6.
- 54 Bender NK, Heilig CE, Dröll B et al. Immunogenicity, efficacy and adverse events of adalimumab in RA patients. Rheumatol Int 2007; 27:269–74.
- 55 Gottlieb A, Menter A, Mendelsohn A et al. Ustekinumab, a human interleukin 12/23 monoclonal antibody, for psoriatic arthritis:

randomised, double-blind, placebo-controlled, crossover trial. Lancet 2009; **373**:633–40.

- 56 Sandborn WJ, Gasink C, Gao LL et al. Ustekinumab induction and maintenance therapy in refractory Crohn's disease. N Engl J Med 2012; 367:1519–28.
- 57 Rojas JR, Taylor RP, Cunningham MR et al. Formation, distribution, and elimination of infliximab and anti-infliximab immune complexes in cynomolgus monkeys. J Pharmacol Exp Ther 2005; 313:578–85.
- 58 Maini RN, Breedveld FC, Kalden JR et al. Therapeutic efficacy of multiple intravenous infusions of anti-tumor necrosis factor α monoclonal antibody combined with low-dose weekly methotrexate in rheumatoid arthritis. *Arthritis Rheum* 1998; **41**:1552–63.
- 59 Vermeire S, Noman M, Van Assche G et al. Effectiveness of concomitant immunosuppressive therapy in suppressing the formation of antibodies to infliximab in Crohn's disease. Gut 2007; 56:1226–31.
- 60 Weinblatt ME, Keystone EC, Furst DE et al. Adalimumab, a fully human anti-tumor necrosis factor α monoclonal antibody, for the treatment of rheumatoid arthritis in patients taking concomitant methotrexate: the ARMADA trial. *Arthritis Rheum* 2003; **48**:35–45.
- 61 Zachariae C, Mork NJ, Reunala T et al. The combination of etanercept and methotrexate increases the effectiveness of treatment in active psoriasis despite inadequate effect of methotrexate therapy. *Acta Derm Venereol* 2008; **88**:495–501.
- 62 Maneiro JR, Salgado E, Gomez-Reino JJ. Immunogenicity of monoclonal antibodies against tumor necrosis factor used in chronic immune-mediated inflammatory conditions: systematic review and meta-analysis. JAMA Intern Med 2013; 173:1–13.
- 63 de Vries MK, van der Horst-Bruinsma IE, Nurmohamed MT et al. Immunogenicity does not influence treatment with etanercept in patients with ankylosing spondylitis. Ann Rheum Dis 2009; 68:531–5.
- 64 Arends S, Lebbink HR, Spoorenberg A et al. The formation of autoantibodies and antibodies to TNF- α blocking agents in relation to clinical response in patients with ankylosing spondylitis. Clin Exp Rheumatol 2010; **28**:661–8.